Author's response to reviews

Title: Acceptability and feasibility of integrating Tuberculosis Screening and Detection in post natal services

Authors:

Charity Mrs Ndwigga CN (cndwiga@popcouncil.org)
Harriet Dr Birungi HB (hbirungi@popcouncil.org)
Chi-Chi Dr Undie CU (cundie@popcouncil.org)
Herman Dr Wayenga HW (howeyenga@gmail.com)
Joseph Dr Sitienei JS (jsetienei@yahoo.com)

Version: 2 Date: 18 October 2012

Author's response to reviews:

Dear Editor

we have revised the abstract to more shorter and more concise one. Also included in the revision is the authors competing interest as well as the financial interests. we have also edited the paper.

Thanks
Charity